Author’s response to reviews

Title: Characterizing measles transmission in India: a dynamic modeling study using verbal autopsy data

Authors:

Stéphane Verguet (verguet@hsph.harvard.edu)
Ed Jones (edward.jones@lshtm.ac.uk)
Mira Johri (mira.johri@umontreal.ca)
Shaun Morris (shaun.morris@sickkids.ca)
Wilson Suraweera (Suraweera@smh.ca)
Cindy Gauvreau (Cindy.Gauvreau@partnershipagainstcancer.ca)
Prabhat Jha (JhaP@smh.ca)
Mark Jit (Mark.Jit@lshtm.ac.uk)

Version: 3 Date: 30 Jun 2017

Author’s response to reviews:

Subject: “Characterizing measles transmission in India: a dynamic modeling study using verbal autopsy data” by Stéphane Verguet, Edward O. Jones, Mira Johri, Shaun K. Morris, Wilson Suraweera, Cindy L. Gauvreau, Prabhat Jha, and Mark Jit

Dear Editors,

In detail below, we give answers to the comments raised in the review. Thank you very much for your consideration of our manuscript.

Sincerely,

Stéphane Verguet
Stéphane Verguet, MS, MPP, PhD  
Assistant Professor of Global Health  
Harvard T.H. Chan School of Public Health

Editorial comments

Please add statement that study approval and ethics were requested or not needed.

Authors’ response: we have added a corresponding statement. Please see Methods section (page 6) and supplementary web appendix (page 23).

Please add a list of abbreviations.

Authors’ response: we have now added a list of abbreviations. Please see “List of abbreviations” (page 3).

Reviewer #1: I am satisfied with the Authors' explanation of the simulation experiment and happy to recommend this manuscript for publication.

Authors’ response: we thank Reviewer #1.